Author's response to reviews

Title: Ileo-sigmoid fistula and delayed ileal obstruction secondary to blunt abdominal trauma: a case report.

Authors:

Konstantinos Bouliaris (kwstisbool@yahoo.com)
Dimos Karangelis (dimoskaragel@yahoo.gr)
Konstantinos Spanos (kostasspanos@meduth.gr)
Stylianos Germanos (sgermn@meduth.gr)
Evangelos Alexiou (ealexiou@meduth.gr)
Anargyros Giaglaras (giagh@yahoo.gr)

Version: 4 Date: 23 July 2011

Author's response to reviews:

Dear editor,

Thank you very much for giving us the opportunity to revise our manuscript. In the revised one we completed all the necessary changes that were requested.

More specifically, we corrected the word legends in the section of figure legends as per reviewer requests and we also modified in the same section the description of figure one. All changes are now underlined. Once again thank you very much for the flawless cooperation so far. Looking forward to receiving your response.

Kindest regards,

Dimos Karangelis MD, PhD